5630 THE SEPARATION OF OWNERSHIP AND CONTROL

Stephen G. Marks

Professor, Boston University School of Law
© Copyright 1999 Stephen G. Marks

Abstract

The separation of ownership and control refers to the phenomenon associated with publicly held business corporations in which the shareholders (the residual claimants) possess little or no direct control over management decisions. This separation is generally attributed to collective action problems associated with dispersed share ownership. The separation of ownership and control permits hierarchical decision making which, for some types of decisions, is superior to the market. The separation of ownership and control creates costs due to adverse selection and moral hazard. These costs are potentially mitigated by a number of mechanisms including business failure, the market for corporate control, the enforcement of fiduciary duties, corporate governance oversight, managerial financial incentives and institutional shareholder activism.

JEL classification: K22

Keywords: Separation of Ownership and Control, Corporate Law, Market for Corporate Control, Corporate Governance

1. Introduction

The separation of ownership and control refers to the phenomenon associated with publicly held business corporations in which the shareholders (the residual claimants) possess little or no direct control over management decisions. Reference to the separation of ownership and control, and concern over its effect, go back at least to Adam Smith. In *The Wealth of Nations*, Smith (1776), writing about joint stock companies, stated:

The directors of such companies ..., being the managers rather of other people's money than of their own, it cannot well be expected that they should watch over it with the same anxious vigilance with which the partners in a private copartnery frequently watch over their own. Like the stewards of a rich man, they are apt to consider attention to small matters as not for their master's honour, and very easily give themselves a dispensation from having it. Negligence and profusion, therefore,

must always prevail, more or less, in the management of the affairs of such a company. It is upon this account that joint stock companies for foreign trade have seldom been able to maintain the competition against private adventurers. (Book 5, Chapter 1, Part 3, Art. 1)

Modern interest in the separation of ownership and control is often associated with Berle and Means (1932) who documented the rise of the modern corporation in the United States. This chapter is organized as follows. Section 2 discusses the legal and economic factors that separate ownership and control. Section 3 lays out the economic benefits of this separation. Section 4 outlines the costs of the separation of ownership and control. Section 5 surveys the various mechanisms for mitigating the costs of the separation of ownership and control. Section 6 considers briefly the origins of these mechanisms. Section 7 summarizes. The examples and motivations will be taken from corporate law in the United States. This chapter will lay out some of the important issues and point the reader to some of the literature in this area as well as give some examples of the economic reasoning involving some of the issues.

2. Ownership and Control

We can define the separation of ownership and control with reference to the owner managed firm. In such a firm, the owner/manager possesses two principal attributes. The owner/manager (1) makes management decisions of the firm and (2) has a claim to the profits of the firm. (These claims are sometimes called residual claims to reflect that they accrue after all costs and fixed claims have been satisfied.) In a large publicly-held corporation, the shareholders own residual claims but lack direct control over management decision making. Correspondingly, managers have control but possess relatively small (if any) residual claims.

The lack of control by shareholders is generally attributed to what is variously called free-rider, collective action or coordination problems. Shareholders in a publicly held corporation typically have limited legal rights to control the corporation. Shareholders do not have the right to engage in any day-to-day management of the corporation. Nor are they able to direct policy or to set compensation. And although shareholders have the right to elect directors, the management controls the voting (proxy) machinery. Typically in the US, management may use corporate funds to solicit proxies while insurgents may use corporate funds only if successful. In spite of this, voting could be an effective instrument of control absent collective action problems. However, the existence of collective action problems greatly weakens and, in many cases, eliminates voting as an effective control mechanism. In order for

a shareholder to oust current management such a shareholder faces significant expected costs. The expected return on such an investment to the shareholder, however, is a small fraction of the total return and, more importantly, small relative to the costs incurred.

As an example, suppose that a shareholder possesses one-hundredth of 1 percent of a corporation's stock. Suppose that a change in management would be worth \$10,000,000 to the corporation and thus \$1000 to the shareholder. Let us suppose that the cost of a proxy solicitation is \$100,000. Given these costs, a shareholder would be unwilling to engage in a proxy solicitation unless almost certain of winning. (In such a case, the expenses would be reimbursed.) What is the chance of winning? Consider the collective action problems facing the solicited shareholder. The likelihood that an individual shareholder's vote will affect the outcome is so small that most shareholders will not take the considerable time and effort of reading and understanding the competing proxy solicitations. (This phenomenon is sometimes called rational apathy.) The result is that the chance that a challenger will win with even a meritorious position, is small. If the chances are, say, 10 percent then the expected gain to the would-be challenger is now \$100. The expected costs are 90 percent the chance of failure) times the cost of solicitation. In our example, this expected cost is \$90,000. Thus shareholders typically have little incentive to engage in such challenges even where the total expected gain of such an action is large relative to the costs.

3. The Benefits of the Separation of Ownership and Control

The benefits of separating ownership and control come from the interaction of three factors. First, under certain conditions and for certain types of decisions, hierarchical decision making may be more efficient than market allocation. Second, due to economies of scale in both production and decision making, optimal firm size can be quite large. Third, optimal investment strategy requires investors to be able to diversify and pool and to be able to change their allocations in response to changing market conditions.

Under some conditions, hierarchical decision making may be more efficient than market transactions. Both hierarchical structures and market structures impose transaction costs. For some types of transactions, market costs may be particularly high. If so, then hierarchical decision making may be more efficient. (see, among others, Coase, 1937 and Williamson, 1979).

As an example, suppose that a firm is considering either purchasing or making an input product. Let us consider two types of costs. Production costs refer to the cost of making the good or to the purchase price if the good is purchased from an outside supplier. Transaction costs refer to the costs of negotiating the transaction. If the input good is a standardized good then it

makes sense for a firm to go to the market and simply buy the input. Since the good is standardized, production costs will be much lower to outside producers who already are producing the good than it will be to produce in-house. In addition, transaction costs will be lower through the market. The purchasing manager simply has to check a couple of prices and then make the purchase. The competitive market will constrain prices and make extensive negotiations unnecessary. In-house production, on the other hand, involves significant transaction costs including communication costs and agency costs. Thus it makes no sense to make the product in-house.

Instead, suppose that the input product is a highly idiosyncratic custom design. Suppose that, because of the uniqueness of the input product, it will cost about the same to build the product in-house as to have someone else build it. In addition, the transaction costs of building it in-house are likely to be lower. Because of the idiosyncratic nature of the product, the amount of negotiation and dispute resolution between a buyer and an outside supplier will be very large. If the product is made in-house, management can communicate to the engineers and the production people what they want. Later, if redesign is necessary, this too can be ordered. (Such redesign negotiations with outside suppliers can be long and tense and can lead to disputes over pricing, delivery dates, and so on.) Although there are principal-agent problems associated with hierarchical decision making within the firm, being able to make command decisions may be much less time-consuming than the extensive negotiations required when dealing with an idiosyncratic product and outside suppliers. Since production costs are the same and the transaction costs are less, the company will likely decide to make the product rather than buy it on the market. (For a more detailed discussion of idiosyncrasy and contractual governance, see Williamson, 1979.)

For many firms, the number of idiosyncratic inputs is large. An organizational form that combines hierarchical organization and large size with diversified and liquid investment may possess informational, transactional and productive efficiencies that make it superior to other organizational forms. The separation of ownership and control permits the existence of an organization with these characteristics. In cases where the benefits outweigh the costs (including agency costs) we would expect to see organizations in which ownership and control are separated (see also, Knight, 1921; Arrow, 1974; Chandler, 1977 and Fama and Jensen, 1983).

4. The Cost of the Separation of Ownership and Control

Much attention has been focused on the costs of the separation of ownership and control and policies that do or might reduce these costs. In this section we will consider the framework for examining these costs and these policies. Section 5 will consider the costs and policies specifically.

Adam Smith considered the separation of ownership and control to be problematic in that managers of such companies would lack the incentives to operate the corporation in the same manner as owner-managers and would thus operate the business in an inefficient manner. Following Adam Smith, Jensen and Meckling (1976) characterized the separation of ownership and control as an agency problem. In the agency approach, shareholders are modeled as principals and managers are modeled as agents. Agents, in this model, maximize personal utility. The issue is how to provide the agent with incentives to induce behavior beneficial to the principals, the shareholders. Agency analysis studies the costs of providing such incentives and the costs resulting from the extent to which agents will still deviate from the interests of the principal even in the presence of such incentives. The costs of the separation of ownership and control are thus the usual principal-agent costs: the monitoring expenditures by shareholders, the bonding expenditures by managers and the residual loss from the divergence of behavior (even with monitoring and bonding) from the ideal.

Agency analysis was important first step for the study of the separation of ownership and control. Agency analysis treated managers as economic actors with utility functions distinct from those of their principals and thus formalized Adam Smith's concern over managerial behavior. Agency analysis focused on incentive schemes by which managers' objective functions could be more closely aligned with those of the principals. Such schemes require the ability to monitor and to reward performance.

Ultimately, however, agency analysis proved to be an imperfect fit for the study of the separation of ownership and control. As Stiglitz (1989) has stated, the problem addressed by the principal-agent literature is 'how one individual, the principal ... can design a compensation system ... which motivates another individual, his agent ... to act in the principal's interest'.

Embodied in this statement are two concepts that are at odds with current analyses of the separation of the ownership and control. First, modern analyses do not take as its ideal the notion that the shareholder should have the ability to monitor or control management. Indeed, policies that encourage shareholder control may undermine the benefits of the separation of ownership and control outlined above. Rather, much modern analysis has focused how actors *other than shareholders* may effectively monitor and constrain managerial behavior. These other actors include other stakeholders (such as bondholders), lawyers (in the prosecution of derivative and class action suits), regulatory authorities

(such as the SEC in the United States) and market participants (such as potential acquirers).

Second, and perhaps more importantly, many modern analyses do not assume that it is socially desirable for managers to act in the best interests of their current principals. This is made clear in the following example. Suppose that a publicly-traded company has made an ore discovery. Immediate announcement of the discovery will increase share value by 10 percent. However, if the corporation delays its announcement by eight months, allowing it to buy up property surrounding the ore discovery site, then the share price will double. Finally suppose that this is an actively traded corporation and that it is certain that the shareholders eight months from now will be different from the current shareholders. If we were to require that management act as agents of current shareholders, an immediate announcement would be required. Most modern analyses would reach a different result (see below).

Berle and Means (1932) suggested treating shareholders as investors who have no necessary claims to control:

On the one hand, the owners of passive property, by surrendering control and responsibility over the active property, have surrendered the right that the corporation should be operated in their sole interest ... Neither the claims of ownership nor the those of control can stand against the paramount interests of the community.

Under this framework shareholders are just another set of investors. The issue thus becomes one of determining socially desirable policies for the governance of the corporation. The goals of such policies could include encouraging the efficient flow of investment funds into production and the efficient production and distribution of goods and services. This, in turn, requires a consideration of shareholder protection and looks to policies that provide managers with optimal incentives. Under this model, managers could have duties toward shareholders, but not exactly the same duties that would be suggested by the agency model. Such a model could also include considerations of social justice and could require duties towards other constituencies.

As an example of an approach that eschews the strict agency framework, consider the argument by Easterbrook and Fischel (1982) that managerial resistance to takeovers is socially counterproductive. Under a strict agency approach management should resist takeovers in situations where resistance and subsequent bargaining would lead to a higher premium for the corporation's shareholders. Easterbrook and Fischel argue that even if resistance produces a higher premium for the shareholders the result is socially inefficient. Gains to the target's shareholders are offset by losses to the acquirer's shareholders. In addition there are a number of deadweight losses

such as those flowing from bargaining breakdowns, dilution of managerial incentives for good management and discouragement of search.

Analysis of the costs of the separation of ownership and control is a several step process involving (1) the articulation of societal goals, (2) the determination of how managerial behavior affects those goals and (3) the evaluation of institutional arrangements in terms of how they affect managerial behavior and at what cost. Broadly speaking, there are two reasons why managerial behavior might not conform to the ideal. The first is that managers might not have the incentives to do so. This is sometimes called the moral hazard problem. The second is that managers may not have the ability to do so (that is, managers may be incompetent). This is sometimes called the adverse selection problem. (see, for example, Ayres and Crampton, 1994 and Smith, 1996). In the following section I will consider some of the institutional mechanisms that mitigate the costs of the separation of ownership and control.

5. Mechanisms for Mitigating the Costs of the Separation of Ownership and Control

The separation of ownership and control gives rise to costs in that managers may act in ways that are inefficient or antisocial. Scholars have explored a number of mechanisms that might give managers an incentive to better conform their behavior to the ideal and that might weed out incompetence. Many (but not all) of these mechanisms rely on actors other than the shareholders. I will group these mechanisms into the following six categories: (1) business failure, (2) the market for corporate control, (3) managerial duties, (4) direct managerial financial incentives, (5) corporate governance oversight and (6) shareholder empowerment. In what follows, I give a brief account of some of the issues and analyses that have arisen in the literature surrounding each mechanism.

5.1 Business Failure

Even in the absence of any other mechanism we would expect business failure to eliminate incompetent managers and thus to mitigate problems of adverse selection. We might also expect business failure to constrain moral hazard. However, absent additional mechanisms to control moral hazard, a market that depended solely on business failure might simply eliminate certain types of otherwise efficient organizations. If managers are unable to credibly commit to behavior that benefits residual claimants, these claimants may be unwilling to make investments in organizations in which they lack direct control. As Akerlof (1970) has demonstrated, asymmetric information can result in the wholesale elimination of markets. In other words, absent other control mechanisms, large publicly owned corporations might not be viable.

In modern corporations, other mechanisms do exist to mitigate the problems of adverse selection and moral hazard. These are discussed in the following sections. Some of these mechanisms operate in conjunction with business failure. For example, managers of failing businesses may be replaced by outside directors or through the market for corporate control. This threat of removal mitigates the problem of moral hazard and removal itself mitigates the problem of adverse selection long before the corporation is actually allowed to fail.

5.2 The Market for Corporate Control

Some scholars have argued that if a market for corporate control is allowed to function, directors will be forced to take action to maximize share value or risk a takeover and the resultant loss of job. The proper functioning of such a market reduces the costs of the separation of ownership and control because, even though shareholders lack direct control, managers will be forced by market pressures to act to maximize share price. In addition, incompetent managers will be removed through the takeover process. In this manner, the market for corporate control mitigates both moral hazard and adverse selection problems. (see, among others, Manne, 1965; Butler, 1989; Easterbrook and Fischel, 1991; Romano, 1992; Macey and Miller, 1995).

Two interesting variants of this story are related to dividend policy and capital structure. Some scholars argue that a corporate policy of paying steady dividends, in spite of adverse tax consequences, signals to the market that the corporation will not squander free cash flows. The market recognizes this signal and elevates stock price. (Presumably, firms that did not engage in this practice would suffer from lower stock price and be subject to a takeover; see, for example Easterbrook, 1984a). Other scholars have suggested that capital structure can provide the same sorts of signals. A capital structure that is weighted towards debt creates an obligation on the part of management to pay out future cash flows. Again this reflects a commitment by management not to squander future cash flows. This commitment is valued by the market and is reflected by higher share prices. Corporations that do not make this commitment have depressed share prices and are subject to takeover. The takeover of corporations in leveraged transactions may represent the market for corporate control operating to impose these debt constraints (see, for example, Jensen, 1986).

The notion that capital markets can help reduce costs of the separation of ownership and control is parasitic on the notion that managerial performance is reflected in stock price. The idea that stock prices reflect information is known as the Efficient Capital Market Hypothesis (ECMH) and it comes in three flavors. Weak-form efficiency occurs when stock prices reflect all past stock price information (if the market is weak-form efficient, then technical analysis is unavailing). Semi-strong-form efficiency occurs when stock prices

reflect all public information (if the market is semi-strong-form efficient then both technical and fundamental analysis is unavailing). Strong-form efficiency occurs when stock prices reflect all public and private, information. Empirical evidence seems to suggest that capital markets are semi-strong form efficient in the sense that it is not possible to make abnormal returns through either technical or fundamental analysis (see Brealey and Myers, 1991; Ross et al., 1993).

Critiques of the market for corporate control as a mechanism for mitigating the costs of the separation of ownership and control have had to deal with the efficient market hypothesis. The critiques come in a variety of forms. Note that in order for the market for corporate control to be effective that (1) managerial incompetence or misbehavior must be correlated with takeover activity, and in order for this to occur it is necessary that (2) that there be a correlation between managerial incompetence or misbehavior and stock price.

Various critiques of the market for corporate control attack one or both of the above correlations. For example, one critique attacks the notion that managerial behavior and stock price are highly correlated. Even if stock prices reflect fundamental values, the determinants of these values contain a large random element (that is, random events unrelated to managerial performance are a significant factor in success). This random element weakens the correlation between managerial performance and stock price.

Another critique allows that while the efficient market hypothesis may be correct in that it prevents the extraction of abnormal profits, it nonetheless does not imply that stock prices reflect the fundamental values of the firm. For example, noise traders could randomly cause stocks to diverge from fundamental value. Again, this random element weakens the correlation between managerial performance and stock price (see, among others, Black, 1986 and Stout, 1988).

Another set of critiques attacks the correlation between managerial activity and takeover activity. Under this critique, takeovers are the result not only of managerial incompetence or misbehavior but also of synergies, the accumulation of market power, expropriation (of creditors or of the government, *vis à vis* taxes, or of labor), of empire-building, and so on. All of these other reasons weaken the correlation between managerial performance and takeover activity. For evaluations of possible determinants of takeover activity, including some empirical studies, see, among others, Kaplan (1989), Lee (1992), Ippolito and James (1992) and Romano (1992).

Other critiques focus on the observability of managerial performance by the market participants. If managerial performance is not observable, or is systematically misobserved, then the market is unable to correctly correlate managerial performance and stock price. One such critique focuses on the informational asymmetry between the market and management. Managers may simply have better information than the market but may have difficulty in conveying this message to the markets. A version of this story is known in the

literature as 'signal jamming' and involves the classic prisoner's dilemma (see Fudenberg and Tirole, 1986). Consider a manager who has the choice between Strategy L (long term) and Strategy S (short term). Neither strategy is certain. Associated with each strategy are two outcomes, one better than the other, and two probabilities:

Strategy S (Short term):

Outcome S1 (0.2 probability): Short-term earnings = 10, Total discounted earnings = 60

Outcome S2 (0.8 probability): Short-term earnings = 20, Total discounted earnings = 80

Strategy L (Long term):

Outcome L1 (0.5 probability): Short-term earnings = 10, Total discounted earnings = 80

Outcome L2 (0.5 probability): Short-term earnings = 20, Total discounted earnings = 100

In this example, Strategy L is clearly superior (the worst outcome under Strategy L is equal to Strategy S's best outcome). Suppose, however, that the market cannot observe the chosen strategy or the expected total discounted earnings flowing therefrom. Rather, only current earnings are observable in the short term. Management chooses a strategy, short-term earnings are revealed, and, on the basis of these short-term earnings, the market values the shares. The market uses the short-term earnings as a signal for total discounted earnings.

The signal the market uses depends on what firms in the market typically do. As an example, suppose that every other firm picks Strategy S. In this case, the market will associate short-term earnings of \$20 with total discounted earnings of \$80, and thus bid the short-term share price to \$80. The market will associate short-term earnings of \$10 with total discounted earnings of \$60, and thus bid the short-term share price to \$60. The manager in this environment faces the following choice:

Strategy S (Short term):

Outcome S1 (0.2 probability): Short-term earnings = 10, Market valuation = 60

Outcome S2 (0.8 probability): Short-term earnings = 20, Market valuation = 80

Expected Market Valuation = 76

Strategy L (Long term):

Outcome L1 (0.5 probability): Short-term earnings = 10, Market valuation = 60

Outcome L2 (0.5 probability): Short-term earnings = 20, Market valuation = 80

Expected Market Valuation = 70

Note that the market is valuing solely on the short-term earnings signal. Instead, if every other firm adopted strategy L, the signals would be different:

Strategy S (Short term):

Outcome S1 (0.2 probability): Short-term earnings = 10, Market valuation = 80

Outcome S2 (0.8 probability): Short-term earnings = 20, Market valuation = 100

Expected Market Valuation = 96

Strategy L (Long term):

Outcome L1 (0.5 probability): Short-term earnings = 10, Market valuation = 80

Outcome L2 (0.5 probability): Short-term earnings = 20, Market valuation = 100

Expected Market Valuation = 90

This leads to the following prisoner's dilemma.

Every other firm:

		Strategy S	Strategy L
Our Firm:	Strategy S	76	96
	Strategy L	70	90

The dominant strategy for the firm, indeed for all firms, is to adopt the short-term strategy. It is interesting to note that in equilibrium, with all firms adopting the short-term strategy, the market indeed values shares correctly. In equilibrium, the market is not being fooled. Yet even though the market is not being fooled, and cannot be fooled in the long term, managers still have the incentive to adopt short-term strategies. In this model, a market for corporate control would drive managers to make inefficient decisions.

As another example, consider a two-period model in which the present-value discounted revenues are as follows:

Strategy S (Short term): PV of Period 1 revs = 100, PV of Period 2 revs = 100

Strategy L (Long term): PV of Period 1 revs = 80, PV of Period 2 revs = 130

Suppose that labor costs in period 1 are fixed by contract at 50 but that the period 2 labor bill will be renegotiated. Suppose that bargaining strengths of labor and management are equal so that second period revenues will be equally split between labor and shareholders. Thus the return to shareholders of Strategy S is 50 in both periods or a total of 100. Returns to shareholders of Strategy L is 30 in the first period and 65 in the second or a total return of 95. Thus share price is maximized by S, even though L produces more value. In this case, even though Strategy S benefits shareholders, it is socially inefficient. Again, the market for corporate control drives managers to make inefficient decisions.

It should be noted that market-induced myopia, though theoretically possible, is difficult to identify empirically due to the difficulty in establishing a benchmark.

5.3 The Specification and Enforcement of Managerial Duties

Some scholars have viewed the establishment of managerial duties legislatively, administratively, judicially, or contractually, and then the establishment of an enforcement mechanism as a means of mitigating the costs associated with the separation of ownership and control (see, for example, Brudney, 1966).

We can divide enforcement regimes into two general categories. The first is enforcement through private rights of action. Typically, in such a system, attorneys are provided incentives, through attorneys' fees in derivative or class action suits, to prosecute cases in which directors violate their duties. Without the provision of attorneys' fees, individual shareholders have little incentive to prosecute such cases themselves since they would bear all of the costs but receive a very small fraction of the benefit. The provision of attorneys' fees solves this problem by creating large incentives for attorneys to prosecute these cases. In effect, such a system provides private attorneys general who operate to enforce managerial duties (see Romano, 1991). This system represents one in which someone other than shareholders (that is, attorneys) provides the enforcement mechanism (the shareholder is just a nominal presence). A number of duties are enforceable in this manner, such as, in the US, the duties of care and loyalty.

The second enforcement mechanism is direct enforcement by the government. In the US, a number of duties are enforceable in this manner, including many duties created through the securities acts. Many duties are enforceable through both private and public enforcement.

The following issues have occupied scholars who have studied this class of mechanisms. What is the optimal set of managerial duties? Should they be broad or narrow? To whom should duties be owed? Should duties be mandatory (state-imposed) or optional (contained in charter provisions)? Should enforcement be private (through derivative and class action suits) or public? Is enforcement effective? Who should be liable, managers, the corporation or both? As to private enforcement, what sort of incentives and disincentives should lawyers have to bring such suits? To what extent should managers be able to immunize themselves through insurance and indemnification? (In addition, there is a question as to the history and origins of the duties and whether the process of producing duties is a good one. I put this last question off to a later section since it relates to corporate law in general and not just to this particular mechanism.)

As to the issue of which duties are best privately enforced and which are best publicly enforced, there is a large literature which is not limited to corporate law. For example, Calabresi and Melamed (1972) and Cooter (1984) both discuss rationales for different enforcement regimes (see also *Boston University Law Review* (1996). In private enforcement, there exists a large literature on optimal incentives to sue which again is not limited to the corporate area (see, for example, Shavell, 1982). Specific to corporate law, there is much debate on whether private enforcement through derivative actions is effective (see, for example, Kraakman, Park and Shavell (1994).

An additional enforcement issue which has been treated by law and economics scholars is whether liability should run to organizations, natural persons or both (see, for example, Arlen, 1994; Arlen and Kraakman, 1997).

One important issue that has been addressed by law and economics scholars is whether duties should be mandatory or optional. Under an optional (opt-out) regime, the law would provide a standard set of duties that would be optimal for a large number of corporations. However, individual corporations would be allowed to opt out of these duties through provisions in the corporate charter. Under this view, the corporation is seen as a nexus of voluntary contracts designed to minimize agency costs. (This nexus of contracts view of the firm probably dates from Coase, 1937. See also Alchian and Demsetz, 1972; Jensen and Meckling, 1976; Klein, 1982; Cheung, 1983; Williamson, 1984; Butler, 1989; Butler and Ribstein, 1990; Easterbrook and Fischel, 1991).

Proponents of the opt-out regime point out that the same set of provisions is unlikely to be optimal for every corporation. They also assert that the existence of relatively efficient markets for securities will create incentives for corporations to adopt provisions that are efficient. Those corporations with inefficient provisions will have depressed share prices and will invite takeovers. As such, the adoption of managerial duties by a corporation is similar to any other managerial decision. (For examples of analyses in support of an opt-out regime, see, Winter, 1977; Carlton and Fischel, 1983; Easterbrook and Fischel, 1982; Macey, 1984; Easterbrook, 1984, among others.)

Proponents of mandatory managerial duties point to market failure. Consider the case for the mandatory provision of information. One argument is that efficient markets depend on the availability of information. Without a duty to disclose there will be an inadequate provision of information. Efficient charter provisions, including efficient disclosure provisions, are unlikely to be produced if the market for securities is inefficient. Thus efficient disclosure provisions are unlikely to be produced in an inefficient market and an efficient market is unlikely to develop absent efficient disclosure provisions (Joseph Heller referred to this type of situation as a Catch-22). Thus, the argument goes, mandatory disclosure provisions are necessary.

A similar argument could be made for provisions (such as poison-pill provisions) that insulate management from the market. Again, efficient takeover provisions are unlikely to be produced in an inefficient market and an efficient market is unlikely to develop absent efficient take-over provisions.

Even if markets are relatively efficient, market forces may provide inadequate incentives. For example, Bebchuk (1989) argues that a charter amendment that redistributes corporate money to management would affect the probability of a takeover only marginally given that such transfers are probably small relative to corporate assets though large relative to individual assets.

Other market failure critiques point to the public-goods nature of managerial duties. This argument asserts that a change in duties could have a large negative impact on the corporation. However, the impact on individual shareholders could be small due to small individual share holdings and thus it would not be worth the cost of resistance. For example, imagine a relaxation of the duty of loyalty. Such a change could result in large benefits for the manager by allowing the manager to expropriate a percentage of corporate wealth. Since the change is small to individual shareholders, such shareholders would not bother to resist. Share price could fall representing the amount that managers have taken. Such an expropriation could not be remedied through the market for corporate control since corporate wealth has already been removed from the corporation.

Yet other critiques point to the lack of sophistication of shareholders. The previous story is enhanced, for example, if the change in the charter is relatively complex. An unsophisticated shareholder would then have to invest more heavily in order to understand and resist such a change. Another critique of the opt-out duties, attacks the wealth-maximization criterion. Under this critique, even if opt-out duties did result in efficiency, mandatory duties would be necessary to accomplish other societal goals such as distributive goals. Several empirical studies have attempted to determine whether corporate decisions to opt out of managerial duties are efficient. (For analyses in support of a mandatory regime see Bebchuk, 1989; Brudney, 1985, and Coffee, 1988, among others.)

5.4. Corporate Governance Structure and Oversight

Under this rubric I include policies directed at the structure and procedures of the board of directors. Some important questions are: How should the board be selected? Who should be on the board of directors? Who should the board represent? What should be the goal of the board? How should the board be compensated?

Boards of directors in US corporations are typically elected by shareholders. In Europe, there has been some experience with boards that represent other constituencies, such as labor. In addition, large share-holdings by banks in countries such as Germany and Japan have provided oversight by lenders. A body of literature examines the costs and benefits of these various representational schemes (see, for example, Aoki, 1984; Jensen and Meckling, 1979; Summers, 1980, and Williamson, 1984). Additionally there exists

literature as to whether boards, however chosen, have responsibilities narrowly to shareholders or broadly to the society as a whole. For example, Dodd (1932) has argued that managers have responsibilities to the community and not just to shareholders. Friedman (1970), on the other hand, has argued that the manager has a duty to maximize shareholder value subject only to legal and ethical constraints. Recently, the debate has been reenergized by the passage in several states of 'other-constituency' statutes. Proponents view these statutes as providing boards with the right to consider other constituencies, such as labor. Opponents view these statutes as merely providing cover for managerial entrenchment.

An additional issue concerns the role of boards in active management and oversight. Some scholars have examined the separation of roles between the chairman of the board (oversight) and the chief executive officer (management). Others have examined the division between outside directors (oversight) and inside directors (management). In the US, boards are typically composed of inside and outside directors. Inside directors manage the day-to-day operations of the corporation. Outside directors provide oversight. Ideally, monitoring by outside directors could mitigate problems of moral hazard and adverse selection (see, for example, Weisbach, (1988).

The reward structure for inside and outside directors differ markedly. For example, in the takeover context, both inside and outside directors stand to lose their jobs. However, whereas for the insider directors this loss is potentially great, including loss of a large income, important responsibilities, prestige and the like, for the outside director this involves the loss of a very part-time job and a very part-time salary. Thus the outside director has diminished incentives to oppose the takeover for personal reasons.

Although outside directors may be more likely to act in the interests of the corporation, the position of outside directors is weakened, however, by the fact that such directors are typically chosen by inside directors and that their continued employment depends on getting along with the insiders. In addition, inside directors and outside directors often find their positions reversed with respect to other corporations. There is thus an incentive for mutual back-scratching. In addition, insider directors typically control the flow of information. This has led some commentators to be skeptical over the efficacy of this mechanism.

Scholars and investors (particularly institutional investors) have made proposals for corporate governance reform that are designed to increase the independence and influence of outside directors. Plans for reform often include one or more of the following proposals: (1) increase the number of outside directors relative to inside directors (most proposals specify that outside directors should be in the majority); (2) remove inside directors from the process of nominating new directors; (3) remove inside directors from the

process of setting directorial compensation; (4) set a mandatory retirement age or term limits for directors; (5) prohibit interlocking directorship (where inside directors of one company are outside directors of another and vice versa, and (6) require that directors own stock in the company. (For additional analysis, see, among others, Mace, 1971; Jensen, 1993, and Lin, 1996.)

5.5 The Alignment of Direct Managerial Financial Incentives

Through derivative financial instruments, particularly warrants, it may be possible to give managers greater incentives to maximize share value. To the extent that share price maximization is desirable this mechanism may reduce the costs associated with the separation of ownership and control. Note, however, that compensatory incentive mechanisms address primarily the problem of moral hazard, not adverse selection.

Imagine, for example, a corporation whose shares are currently trading for \$100. Suppose that at-the-money warrants are trading at \$1. Now imagine the following compensation schemes. Scheme 1 pays the manager \$200,000 cash per year. Scheme 2 pays the manager \$100,000 cash plus \$100,000 worth of restricted shares (that is, 1000 shares). The restricted shares cannot be traded for five years. Scheme 3 pays the manager \$100,000 cash plus \$100,000 worth of warrants (that is, 100,000 warrants). The warrants would be exercisable in five years. Each scheme provides the same value of compensation. However, the incentive effects are quite different. Under scheme 1, the manager is not required to take an equity interest in the corporation. Under scheme 2, the manager will have accumulated 5000 shares after five years. Under scheme 3, the manager will possess 500,000 warrants after five years. Imagine now that a takeover offer for the corporation develops involving a 100 percent premium. Under scheme 1, the manager has an incentive to resist the takeover even if the price is a good one since the manager will likely lose his or her job. Under scheme 2, the manager has something of an incentive to consider the opportunity of a good price since the manager will receive \$500,000 in increased value for selling into the offer as opposed to resisting it. Under scheme 3, the share price is likely to dominate the managers thinking since the premium represents \$50,000,000 to the manager. (Assume that the manager may exercise the warrants in a takeover situation.) To the extent that maximization of share price is a desired outcome this mechanism creates stronger incentives for managers than might otherwise exist.

The warrant mechanism could be combined with corporate governance structures in order to create even stronger incentives. For example, one could imagine governance structures in which outside directors constituted a majority and in which outside directors were compensated exclusively through warrants. In such a case, outside directors would have small adverse incentives (since the salary is small) and significant correct incentives since their compensation would depend almost exclusive on share price performance.

One should note that warrants create asymmetric incentives. That is, they pay if share price is above the exercise price. However, the payoffs are the same (that is, zero) for all prices below the exercise price. Thus warrants could create incentives for inefficient risk-taking. Nevertheless, since cash compensation creates incentives for inefficiently risk-averse behavior it is not clear what the net effect of warrants would be (see, among others, Jensen and Murphy, 1990; Gilson, 1992; Milgrom and Roberts, 1992).

5.6 Shareholder Empowerment

A number of scholars have made proposals to increase the power of shareholders over management. Shareholder empowerment as a mechanism for managerial control has a different emphasis from the previously discussed mechanisms in that those mechanisms relied on actors other than shareholders to restrain management. Those mechanisms sought to maintain the advantages of the separation of ownership and control while mitigating the costs. By contrast, shareholder empowerment reduces the separation and gives shareholders more say in management. Thus some of the critiques of these proposals point to the compromising of centralized management and reduction of the benefits that flow therefrom (see, for example, Boyer, 1993).

Shareholder empowerment proposals seek to give public shareholders a greater voice in management by endowing them with greater rights to manage and by reducing the costs of involvement. An important subset of this research notes the increasing prominence of institutional investors in the securities markets and evaluates the costs and benefits of legal changes that would encourage greater activism by institutional investors.

Current proxy regulation in the United States gives incumbent management a tremendous advantage over challengers to their authority. Very few matters are required to be put to the vote of shareholders. Managers can fund proxy battles out of the corporate treasury while challengers must incur the high costs themselves unless they ultimately prove successful. In addition, corporations have the ability to restrict voting rights in certain classes of shares (see, for example, Seligman, 1986). This system, combined with collective action problems (see Section 2 above), discourages challenges even where the total benefit of the challenge to the corporation is great compared to the costs. Some proposals for corporate reform would alter proxy rules to put more decisions in the hands of shareholders (that is, to provide for binding shareholder resolutions) and to change the cost structure to make challenges significantly cheaper, for example, by putting some of the resources of the corporation into the hands of shareholder challengers (see, among others, Bernstein and Fischer, 1940; Caplin, 1953; Eisenberg, 1970; Dent, 1989; Goforth, 1994; Smith, 1996). Evaluation of such proposals must trade the gains of greater managerial accountability against both the costs of increased use of corporate funds for such purposes and the cost of decreased centralization of management

decisionmaking.

In the US, equity securities are held increasingly by institutions, i.e., pension funds, insurance funds and investment funds. Although any individual fund holds a small percentage of the shares of any given corporation, the holdings are significant enough to encourage some monitoring. In addition, a small number of funds could control enough of a corporation to possess significant voting power if their actions are coordinated (see, among others, Gilson and Kraakman, 1991, 1993a; Black, 1992a, 1992b; and Maug, 1997). In the US, laws exist which work to discourage cooperation among funds to exercise voting control. Some proposals have suggested reducing these barriers to cooperation in order to allow funds to exert control over management and thus to reduce the separation of ownership and control and thereby reduce the costs associated with the separation of ownership and control. Some of this scholarship has been comparative in nature, focusing on institutional ownership by banks in countries such as Germany and Japan (see, among others, Gilson and Roe, 1993).

Skeptics of institutional monitoring point to a number of costs associated with institutional investor oversight. Fund managers are themselves agents whose interests are not aligned with their own investors. In addition, managers represent only some of the shareholders. Control and access to information may lead to insider trading and other abuses *vis-à-vis* other shareholders. Although fund managers may have incentives to cooperate with other fund managers in order to exercise control, they are also in competition. Strategic behavior may result in inefficient control. Some managers, particularly managers of government pension funds, may be susceptible to political influence. Debates over institutional investor voice involve assessing these costs (see, among others, Garten, 1992; Romano, 1993; Calio and Zahralddin, 1994; Bainbridge, 1993, 1995; Fisch, 1994; Utset, 1995).

6. Origins of the Mechanisms for Mitigating the Costs of the Separation of Ownership and Control

In the United States, most of the mechanisms for mitigating the costs of ownership and control are regulated by state law. Some commentators have suggested that the optimal development of mechanisms has been impeded through the process of state competition for corporate charters. This has become known as the 'race-to-the-bottom' thesis. This has led some commentators to support a federal (that is, national) chartering of corporations and the development of a federal corporate law. Other commentators argue that jurisdictions that compete for charters will offer efficient corporate laws. Yet other scholars have applied public choice tools to explain the development of corporate law as the result of the interest group interaction (see Cary, 1974;

Dodd and Leftwich, 1980; Romano, 1985, 1994; Hovenkamp, 1988; Macey and Miller, 1987; Bebchuk, 1992; Bratton, 1994; and Ayres, 1995).

7. Summary

Many of the issues in corporate law can be traced to the separation of ownership and control. This separation has made viable large, centrally-managed firms with diversified and liquid investors. It has provided the means for the efficient mobilization of large amounts of private capital towards productive uses. The same separation of ownership and control has created potential costs. Much of corporate law can be explained as mechanisms designed to mitigate these costs. Economic analysis of law, with its emphasis on incentive effects and on the costs and benefits that flow therefrom, provides an important tool for the analysis of corporate law and policy.

Selected Bibliography on the Separation of Ownership and Control (5630)

- Adams, Walter and Brock, James W. (1990), 'Efficiency, Corporate Power, and the Bigness Complex', 21 Journal of Economic Education, 30-50.
- Agrawal, Arun and Knoeber, Charles R. (1996), 'Firm Performance and Mechanisms To Control Agency Problems Between Managers and Shareholders', 31 Journal of Financial and Quantitative Analysis, 377-397.
- Agrawal, Arun and Mandelker, G.N. (1987), 'Managerial Incentives and Corporate Investment and Financing Decisions', **42** *Journal of Finance*, 823-837.
- Agrawal, Arun and Mandelker, G.N. (1990), 'Large Shareholders and the Monitoring of Managers: The Case of Antitakeover Charter Amendments', 25 Journal of Financial and Quantitative Analysis, 143-161.
- Agrawal, Arun and Nagarajan, N.J. (1990), 'Corporate Capital Structure, Agency Costs and Ownership Control: The Case of All-equity Firms', **45** *Journal of Finance*, 1325-1331.
- Agrawal, Arun and Walkling R. (1994), 'Executive Careers and Compensation Surrounding takeover Bids', **49** *Journal of Finance*, 985-1014.
- Agrawal, Arun, Makhija, A.K. and Mandelker, G.N (1991), 'Executive Compensation and Corporate Performance in Electric and Gas Utilities: Some Empirical Evidence', 20 Financial Management, 113-124.
- Akerlof, George A. (1970), 'The Markets for 'Lemons': Qualitative Uncertainty and the Market Mechanism', **84** *Quarterly Journal of Economics*, 488-500.
- Alchian, Armand A. and Demsetz, Harold (1972), 'Production, Information Costs, and Economic Organization', 62 American Economic Review, 777-795.
- Alchian, Armen A. (1965), 'The Basis of Some Recent Advances in the Theory of Management of the Firm', 14 Journal of Industrial Economics, 30-41.
- Alchian, Armen A. (1969), 'Corporate Management and Property Rights', in Manne, Henry G. (ed.), Economic Policy and the Regulation of Corporate Securities, Washington, American Enterprise Institute for Public Policy Research.

Alfaro Aguila-Real, Jesús (1995), Interés Social y Derecho de Suscripción Preferente. Una aproximación económica (Economic Efficiency of a strict Construction of the 'Interest of the Corporation' Clause in Corporate Law), Madrid, Ed. Cívitas, 160 p.

Amihud, Y. and Lev, B. (1981), 'Risk Reduction as a Managerial Motive for Conglomerate Mergers', 12 Bell Journal of Economics, 605-617.

Andre, Thomas J., Jr (1990), 'A Preliminary Inquiry into the Utility of Vote Buying in the Market for Corporate Control', 63 Southern California University Law Review, 533 ff.

Aoki, Masahiko (1984), The Cooperative Game Theory of the Firm, Oxford, Oxford University Press.
 Arlen, Jennifer H. (1994), 'The Potentially Perverse Effects of Corporate Criminal Liability', 23
 Journal of Legal Studies, 833 ff.

Arlen, Jennifer H. and Kraakman, Reinier H. (1997), 'Controlling Corporate Misconduct: An Analysis of Corporate Liability Regimes', 72 New York University Law Review, 687-779.

Arlen, Jennifer H. and Weiss, Deborah (1995), 'A Political Theory of Corporate Taxation', 105 Yale Law Journal, 325 ff.

Arrow, Kenneth J. (1974), The Limits of Organization, New York, Norton.

Arruñada, Benito (1988), 'Un Análisis Económico de la Regulación de la Sociedad anónima en España (An Economic Analysis of the Regulation of the Joint-Stock Society in Spain)', 3 Anales de Estudios Económicos y Empresariales, 191-224.

Arruñada, Benito (1990), Control y Regulación de la Sociedad Anónima (Control and Regulation of Joint-Stock Companies), Madrid, Alianza Editorial.

Arruñada, Benito (1990b), 'Control y Propiedad: Limites al Desarrollo de la Empresa Española (Corporate Control and Ownership: Limits to its Growth in Spain)', 687 Información Comercial Española - Revista de Economia, 67-88.

Arruñada, Benito (1992), 'Critica a la Regulación de las Opas (A Critic on the Regulation of Takeovers)', 203-4 Revista de Derecho Mercantil, 29-67.

Arruñada, Benito (1992b), 'La Conversión Coactiva de Acciones Comunes en Acciones sin Voto para Lograr el Control de las Sociedades Anónimas: De Cómo la Ingenuidad Legal Prefigura el Fraude (Mandatory Conversion of Common Shares into Non-Voting Shares to Reach Corporate Control over Joint-Stock Companies: How legal Simplicity Leads to Fraud)', 71 Revista Española de Financiación y Contabilidad, 283-314.

Arruñada, Benito and Paz Ares, Cándido (1995), 'Conversion of Ordinary Shares into Non-Voting Shares', **15** International Review of Law and Economics.

Ayres, Ian (1991), 'Back to Basics: Regulating How Corporations Speak to the Market', 77 Virginia Law Review, 945 ff.

Ayres, Ian (1995), 'Supply-Side Inefficiencies in Corporate Charter Competition: Lessons From Patents, Yachting and Bluebooks', 43 University of Kansas Law Review, 541 ff.

Ayres, Ian and Crampton, Peter (1994), 'Relational Investing and Agency Theory', 15 Cardozo Law Review, 1033 ff.

Bab, Andrew Laurance (1991), 'Debt Tender Offer Techniques and the Problem of Coercion', 91 Columbia Law Review, 846-890.

Bainbridge, Stephen M. (1993a), 'In Defence of the Shareholder Wealth Maximization Norm: A Reply to Professor Green', 50 Washington and Lee Law Review, 1423 ff.

Bainbridge, Stephen M. (1993b), 'Independant Directors and the ALI Corporate Governance Project', **61** George Washington Law Review, 1034 ff.

Bainbridge, Stephen M. (1995), 'The Politics of Corporate Governance', 18 Harvard Journal of Law

- and Public Policy, 671 ff.
- Bajt, Aleksander (1968), 'Property in Capital and in the Means of Production in Socialist Economics', 11 Journal of Law and Economics, 1-4.
- Bargman, Diedre A. and Cox, Paul N. (1984), 'Reappraising the Role of the Shareholder in the Modern Public Corporation: Weinberger's Procedural Approach to Fairness in Freezeouts', 1984 Wisconsin Law Review, 593 ff.
- Baysinger, Barry D. and Butler, Henry N. (1985), 'Corporate Governance and the Board of Directors: Performance Effects of Changes in Board Composition', 1 Journal of Law, Economics, and Organization, 101-124.
- Baysinger, Barry D. and Zardkoohi, Asghar (1986), 'Technology, Residual Claimants, and Corporate Control', 2 Journal of Law, Economics, and Organization, 339-349.
- Bebchuck, Lucian Arye (1989), 'Limiting Contractual Freedom in Corporate Law: The Desirable Constraints on Charter Amendments', **102** *Harvard Law Review*, 1820-1860.
- Bebchuck, Lucian Arye (1992), 'Federalism and the Corporation: The Desirable Limits on State Competition in Corporate Law', 105 Harvard Law Review, 1435 ff.
- Bebchuk, Lucian Arye and Kahan, Marcel (1990), 'A Framework for Analyzing Legal Policy Towards Proxy Contest', **78** California Law Review, 1071-1135.
- Berle, A.A., Jr and Means, Gardiner C. (1932), *The Modern Corporation and Private Property*, New York, MacMillan.
- Bernstein, Seldon E. and Fischer, Henry G. (1940), 'The Regulation of the Solicitation of Proxies: some Reflections on Corporate Democracy', 7 *University of Chicago Law Review*, 226 ff.
- Bhagat, Sanjai and Brickley, James A. (1984), 'Cumulative Voting: The Value of Minority Shareholder Voting Rights', 27 Journal of Law and Economics, 339-365.
- Biancala, Joseph (1990), 'Defining the Proper Corporate Constituency: Asking The Wrong Question', **59** *University of Cincinnati Law Review*, 425 ff.
- Bjuggren, Per-Olof (1989), 'Ownership and Efficiency in Companies Listed on Stockhold Stock Exchange 1985', in Faure, Michael and Van den Bergh, Roger (eds), Essays in Law and Economics. Corporations, Accident Prevention and Compensation for Losses, Antwerpen, Maklu. 71-79.
- Black, Bernard (1992a), 'Agents Watching Agents: The Promise of Institutional Investor Voice', 39 UCLA Law Review, 811 ff.
- Black, Bernard (1992b), 'The Value of Institutional Investor Monitoring: The Empirical Evidence', **39** UCLA Law Review, 895 ff.
- Black, Bernard and Kraakman, Reinier H. (1996), 'A Self-Enforcing Model of Corporate Law', 109 Harvard Law Review, 1911-1979.
- Black, Fisher (1986), 'Noise', 41 Journal of Finance, 529 ff.
- Boardman, Anthony E. and Vining, Aidan R. (1989), 'Ownership and Performance in Competitive Environments: A Comparison of the Performance of Private, Mixed, and State-owned Enterprises', 32 Journal of Law and Economics, 1-33.
- Boyer, Allen (1988), 'When it Comes to Hostile Tender Offers, Just Say No: Commerce Clause and Corporation Law in CTS Corp. v. Dynamics Corp. of America', 57 Cincinatti Law Review, 539 ff.
- Boyer, Allen (1993), 'Activist Shareholders, Corporate Directors, and Institutional Investment: Some Lessons from the Robber Barons', **50** Washington and Lee Law Review, 977 ff.
- Bradley, Michael D. and Rosenzweig, Michael (1986), 'Defensive Stock Repurchases and the

- Appraisal Remedy', 96 Yale Law Journal, 322-338.
- Bratton, William W., Jr (1993), 'Confronting the Ethical Case Against the Ethical Case for Constituency Rights', 50 Washington and Lee Law Review, 1449 ff.
- Bratton, William W., Jr (1994), 'Corporate Law's Race to Nowhere in Particular', 44 University of Toronto Law Review, 401 ff.
- Brealey, Richard A. and Myers, Stewart C. (1991), *Principles of Corporate Finance*, New York, McGraw Hill.
- Brickley, James A., Bhagat, Sanjai and Lease, Ronald C. (1985), 'The Impact of Long-range Managerial Compensation Plans on Shareholder Wealth', 7 Journal of Accounting and Economics, 115-130.
- Brudney, Victor (1966), 'Fiduciary Ideology in Transactions Affecting Corporate Control', 65 Michigan Law Review, 259 ff.
- Brudney, Victor (1985), 'Corporate Governance, Agency Costs, and the Rhetoric of Contract', 85 Columbia Law Review, 1403-1444.
- Bundy, Stephen M. (1989), 'Commentary: Rational Bargaining and Agency Problems', 75 Virginia Law Review, 335-365.
- Butler, Henry N. (1989), 'The Contractual Theory of the Corporation', 11 George Mason Law Review, 99 ff
- Butler, Henry N. and Ribstein, Larry E. (1990), 'Opting Out of Fiduciary Duties: A Response to the Anti-Contractarians', **65** Washington Law Review, 1 ff.
- Butler, Henry N. and Ribstein, Larry E. (1994), 'Corporate Governance Speech and the First Amendment', 43 University of Kansas Law Review, 163 ff.
- Calabresi, Guido and Melamed, A. Douglas (1972), 'Property Rules, Liability Rules and Inalienability:
 One View of the Cathedral', 85 Harvard Law Review, 1089-1128. Reprinted in Ackerman, Bruce
 A. (ed.) (1975), Economic Foundations of Property Law, Boston, Little Brown, 31-48.
- Calio, Joseph Evan and Zahralddin, Rafael Xavier (1994), 'The Securities and Exchange Commissions 1992 Proxy Amendments: Questions of Accountability', 14 Pace International Law Review, 459 ff.
- Campbell, David (1990), 'Adam Smith, Farrar on Company Law and the Economics of the Corporation', 19 Anglo-American Law Review, 185-208.
- Campbell, David (1995), 'Review of R. Tomasic and S. Bottomley, Directing the Top 500', 4 Social and Legal Studies, 428-432.
- Campbell, David (1996), 'Review of B. Fisse and J. Braithwaite, Corporations, Crime and Accountability', 5 Social and Legal Studies, 549-554.
- Caplin, Mortimer M. (1953), 'Shareholder Nominations of Directors: A Program for Fair Corporate Suffrage', 39 Virginia Law Review, 141 ff.
- Carlton, Denis and Fischel, Daniel R. (1983), 'The Regulation of Insider Trading', 35 Stanford Law Review, 857 ff.
- Carney, William J. (1988), 'Controlling Management Opportunism in the Market for Corporate Control: An Agency Cost Model', 64 Wisconsin Law Review, 385-433.
- Cary, William L. (1974), 'Federalism and Corporate Law: Reflections Upon Delaware', 83 Yale Law Journal, 663 ff.
- Chandler, Alfred (1977), The Visible Hand, Cambridge, MA, Harvard University Press.
- Charensky, Steven S. (1994), 'Shareholders, Managers, and Corporate R&D; Spending: An Agency Cost Model', 10 Santa Clara Computer and High Tech. Law Journal, 299-346.
- Cheung, Steven N.S. (1983), 'The Contractual Nature of the Firm', 26 Journal of Law and

- Economics, 1-21.
- Choi, Dosoung, Kamma, Sreenivas and Weintrop, Joseph (1989), 'The Delaware Courts, Poison Pills, and Shareholder Wealth'. 5 Journal of Law. Economics, and Organization, 375-393.
- Clarckson, Kenneth W. and Martin, Donald L. (eds) (1979), The Economics of Nonproprietary Institutions (Supplement 1 to Research in Law and Economics), Greenwood, JAI Press, 330 p.
- Clark, William A.V. (1988), 'Production Costs and Output Qualities in Public and Private Employment Agencies', 31 Journal of Law and Economics, 379-393.
- Coase, Ronald H. (1937), 'The Nature of the Firm', 4 Economica, 386-405. Reprinted in Kronman, Anthony T. and Posner, Richard A. (eds) (1979), The Economics of Contract Law, Boston, Little Brown, 31-32.
- Coffee, John C., Jr (1984), 'Market Failure and the Economic Case for A Mandatory Disclosure System', 70 Virginia Law Review, 717-754.
- Coffee, John C., Jr (1986), 'Shareholders Versus Managers: The Strain in the Corporate Web', 85 Michigan Law Review, 1-109.
- Coffee, John C., Jr (1988), 'No Exit?: Opting Out, the Contractual Theory of the Corporation, and the Special Case of Remedies', 53 Brooklyn Law Review, 919 ff.
- Coffee, John C., Jr (1989), 'The Mandatory/Enabling Balance in Corporate Law: An Essay on the Judicial Role', 89 Columbia Law Review.
- Coffee, John C., Jr (1993), 'New Myths and Old Realities: The American Law Institute Faces the Derivative Action', **48** *Journal of Business Law*, 1407 ff.
- Coffee, John C. (1994), Transfers of Control and the Quest for Efficiency: Can Delaware Law Encourage Efficient Transactions While Chilling Inefficient Ones?, Working Paper based on a lecture given on 10/28/94.
- Conard, A.F. (1988), 'Beyond Managerial Capitalism: Investor Capitalism?', 72 University of Michigan Journal of Law Reform, 117-178.
- Cooter, Robert (1984), 'Prices and Sanctions', 84 Columbia Law Review, 1523 ff.
- Cooter, Robert D. and Rubin, Edward L. (1988), 'Orders and Incentives as Regulatory Methods: The Expedited Funds Availability Act of 1987', 35 UCLA Law Review, 1115-1186.
- Cox, James D. (1984), 'Compensation, Deterrence, and the Market as Boundaries for Derivative Suit Procedures', **52** *George Washington Law Review*, 745-788.
- Cutler, David M. and Summers, Lawrence H. (1988), 'The Costs of Conflict Resolution and Financial Distress: Evidence from the Texaco-Pennzoil Litigation', 19 Rand Journal of Economics, 157-172.
- Daines, Robert M. and Hanson, Jon D. (1992), 'The Corporate Law Paradox: The Case for Restructuring Corporate Law, a Review Essay on The Economic Structure of Corporate Law, by Easterbrook and Fischel', 102 Yale Law Journal, 577-637.
- Dallas, Lynne L. (1988), 'Two Models of Corporate Governance: Beyond Berle and Means', 72 University of Michigan Journal of Law Reform, 19-116.
- De Alessi, Louis (1973), 'Private Property and Dispersion of Ownership in Large Corporations', 28 Journal of Finance, 839-851.
- Demsetz, Harold and Lehn, Kenneth (1985), 'The Structure of Corporate Ownership: Causes and Consequences', 93 Journal of Political Economy, 1155-1177.
- Dennis, Roger J. (1987), 'Mandatory Disclosure Theory and Management Projections: A Law and Economics Perspective', 46 Maryland Law Review, 1197 ff.

- Dent, George W., Jr (1989), 'Toward Unifying Ownership and Control in the Public Corporation', 65 Wisconsin Law Review, 881-924.
- Dodd, E. Merrick (1932), 'For Whom are Corporate Managers Trustees?', 45 Harvard Law Review, 1145 ff.
- Dodd, Peter and Leftwich, Richard (1980), 'The Market for Corporate Charters: 'Unhealthy Competition' vs. Federal Regulation', 53 Journal of Business, 1-41.
- Dunbar, Frederick C., Juneja, Vinita M. and Martin, Denise N. (1995), Shareholder Litigation: Deterrent Value, Merit, and Litigants, Options, New York, Law Journal Seminars Press.
- Easterbrook, Frank H. (1984a), 'Two Agency-Cost Explanations of Dividends', 74 American Economic Review, 650-659.
- Easterbrook, Frank H. (1984b), 'Managers' Discretion and Investors' Welfare: Theories and Evidence', **9** *Delaware Journal of Corporate Law*, 540-571.
- Easterbrook, Frank H. and Fischel, Daniel R. (1982), 'Corporate Control Transactions', 91 Yale Law Journal, 698 ff.
- Easterbrook, Frank H. and Fischel, Daniel R. (1983), 'Voting in Corporate Law', 26 Journal of Law and Economics, 395-427.
- Easterbrook, Frank H. and Fischel, Daniel R. (1984), 'Mandatory Disclosure and the Protection of Investors', 70 Virginia Law Review, 669-715.
- Easterbrook, Frank H. and Fischel, Daniel R. (1989), 'The Corporate Contract', **89** Columbia Law Review, 1416 ff.
- Easterbrook, Frank H. and Fischel, Daniel R. (1991), The Economic Structure of Corporate Law, Boston, Harvard Univ Press.
- Eckel, Catherine C. and Vermaelen, Theo (1986), 'Internal Regulation: The Effects of Government Ownership on the Value of the Firm', 29 Journal of Law and Economics, 381-403.
- Eisenberg, Melvin Aron (1970), 'Access to the Corporate Proxy Machinery', **83** Harvard Law Review, 1489 ff.
- Eisenberg, Melvin Aron (1989), 'The Structure of Corporation Law', **89** Columbia Law Review, 1461 ff
- Fama, Eugene F. (1980), 'Agency Problems and the Theory of the Firm', **88** *Journal of Political Economy*, 288 ff.
- Fama, Eugene F. and Jensen, Michael C. (1983a), 'Separation of Ownership and Control', **26** *Journal of Law and Economics*, 301-325.
- Fama, Eugene F. and Jensen, Michael C. (1983b), 'Agency Problems and Residual Claims', 26 Journal of Law and Economics, 327-349.
- Fisch, Jill E. (1994), 'Relationship Investing: Will it Happen? Will it Work?', **55** Ohio State Law Journal, 1009-1048.
- Fischel, Daniel R. (1978), 'Efficient Capital Market Theory, the Market for Corporate Control and the Regulation of Cash Tender Offers', 57 Texas Law Review, 1-46.
- Friedman, Milton (1970), 'The Social Responsibility of Business is to Increase its Profits', 13/9/1970 *The New York Times*.
- Frost, Christopher (1992), 'Running the Asylum: Governance Problems in Bankruptcy Reorganizations', **34** *Arizona Law Review*, 89 ff.
- Fudenberg, Drew and Tirole, Jean (1986), 'A "Signal-Jamming" Theory of Predation', 17 Rand Journal of Economics, 366 ff.
- Garten, Helen A. (1992), 'Institutional Investors and the New Financial Order', 44 Rutgers Law

- Review, 585 ff.
- Garth, Bryant G., Nagel, Ilene H. and Plager, Sheldon J. (1985), 'Empirical Research and the Shareholder Derivative Suit: Toward a Better Informed Debate', 48(3) Law and Contemporary Problems. 137-159.
- Geddes, R. Richard (1997), 'Ownership, Regulation, and Managerial Monitoring in the Electric Utility Industry', 40 Journal of Law and Economics.
- Geddes, R. Richard and Crowley, Peter T. (1994), 'Agency Costs and Governance in the United States Postal Service', in Sidak, Gregory J. (ed.), Governing the Postal Service, Washington, American Enterprise Institute for Public Policy Research.
- Gilson, Ronald J. (1992), 'Executive Compensation and Corporate Governance, An Academic Perspective', **24** Annual Institute on Securities Regulation.
- Gilson, Ronald J. and Kraakman, Reinier H. (1984), 'The Mechanisms of Market Efficiency', 70 Virginia Law Review, 549-644.
- Gilson, Ronald J. and Kraakman, Reinier H. (1991a), 'Corporate Governance and Portfolio Performance: An Agenda for Institutional Investors', *Director's Monthly*, 1-4.
- Gilson, Ronald J. and Kraakman, Reinier H. (1991b), 'Reinventing the Outside Director: An Agenda for Institutional Investors', 43 Stanford Law Review, 863-906.
- Gilson, Ronald J. and Kraakman, Reinier H. (1993), 'Investment Companies as Guardian Shareholders: The Place of the MSIC in the Corporate Governance Debate', 45 Stanford Law Review, 985-1010.
- Gilson, Ronald J. and Roe, Mark J. (1993), 'Understanding the Japanese Keiretsu: Overlaps Between Corporate Governance and Industrial Organization', 102 Yale Law Journal, 871 ff.
- Goforth, Carol (1994), 'Proxy Reform as a Means of Increasing Shareholder Participation in Corporate Governance: Too Little, But Not Too Late', 43 American University Law Review, 379-465.
- Gordon, Jeffrey N. (1989), 'The Mandatory Structure of Corporate Law', 89 Columbia Law Review, 1549-1598.
- Green, Ronald M. (1993), 'Shareholders as Stakeholders: Changing Metaphors of Corporate Governance', 50 Washington and Lee Law Review, 1409-1421.
- Greenwood, Daniel J.H. (1996), 'Fictional Shareholders: For Whom Are Corporate Managers Trustees, Revisited', 69 University of Southern California Law Review, 1021-1104.
- Grossman, Sanford J. and Hart, Oliver D. (1980), 'Takeover Bids, the Free-Rider Problem, and the Theory of the Corporation', 11 Bell Journal of Economics, 42-64.
- Györffy, Gabor (1995), 'The Structure of Fiduciary Relationship: Its Relevance to Nonprofit Governances and Regulation', in Bouckaert, Boudewijn and De Geest, Gerrit (eds), Essays in Law and Economics II: Contract Law, Regulation, and Reflections on Law and Economics, Antwerpen, Maklu, 193-216.
- Hansmann, Henry B. (1988), 'Ownership of the Firm', 4 Journal of Law, Economics, and Organization, 267-304.
- Hansmann, Henry B. (1990), 'When Does Worker Ownership Work? ESOPs, Law Firms, Codetermination, and Economic Democracy', 99 Yale Law Journal, 1749-1816.
- Hindley, Brian (1970), 'Separation of Ownership and Control in the Modern Corporation', 13 Journal of Law and Economics, 185-221.
- Holderness, Clifford G. (1990), 'Liability Insurers as Corporate Monitors', 10 International Review of Law and Economics, 115-129.
- Hovenkamp Herbert J. (1988), 'The Classical Corporation in American Legal Thought', 76 George

- Washington Law Review, 1593-1688.
- Hu, Henry T.C. (1990), 'Risk, Time, and Fiduciary Principles in Corporate Investment', 38 UCLA Law Review. 277-390.
- Hwang, Chun-Sin and Kan, Steven S. (1994), Principles of Economics, Cooperating for Mutual Prosperity and Progress (in Chinese), Published by the authors and distributed by Shin Lu Bookstore, Taipei.
- Hwang, Chun-Sin and Kan, Steven S. (1995), 'Democracy and the Principle for the Division of Labor in Government Organization (in Chinese)', in Chien, Sechin Y.S. and Tai, Terence H. (eds), *Philosophy and Public Norms*, Taipei, Academia Sinica, Sun Yat-sen Institute for Social Sciences and Philosophy, 163-200.
- Ippolito, Richard A. and James, William H. (1992), 'LBOs Reversions and Implicit Contracts', 67 Journal of Finance, 139 ff.
- Itoh, Hideshi (1992), 'Cooperation in Hierarchical Organizations: An Incentive Perspective', 8 Journal of Law, Economics, and Organization, 321-345.
- Janjigian, Vahan and Bolster, Paul J. (1990), 'The Elimination of Director Liability and Stockholder Returns: An Empirical Investigation', 13 Journal of Financial Research, 53-60.
- Jensen, Michael C. (1986), 'Agency Costs and Free Cash Flow, Corporate Finance, and Takeovers', 76 American Economic Review. Papers and Proceedings, 323-329.
- Jensen, Michael C. (1993), 'The Modern Industrial Revolution, Exit, and the Failure of Internal Control Systems', 48 Journal of Finance, 831 ff.
- Jensen, Michael C. and Meckling, William H. (1976), 'Theory of the Firm: Managerial Behaviour, Agency Costs and Ownership Structure', 3 Journal of Financial Economics, 305-360.
- Jensen, M and Meckling, W. (1979), 'Rights and Production Functions: An Application to Labor-Managed Firms and Co-determination', 52 Journal of Business, 469 ff.
- Jensen, Michael C. and Murphy, Kevein M. (1990), 'Performance Pay and Top-Management Incentives', 98 Journal of Political Economy, 225 ff.
- Jensen, Michael C. and Murphy, Kevin J. (1990), 'CEO Incentives It's Not How Much You Pay, But How', 68 Harvard Business Review, 138 ff.
- Johnson, K.C. (1985), 'Golden Parachutes and the Business Judgment Rule: Toward a Proper Standard of Review', 94 Yale Law Journal, 909-928.
- Joskow, Paul L. (1988), 'Asset Specificity and the Structure of Vertical Relationship: Empirical Evidence', 4 Journal of Law, Economics, and Organization, 95-117.
- Kabir, R., Cantrijn, D. and Jeunink, A. (1997), 'Takeover Defenses, Ownership Structure and Stock Returns in the Netherlands: An Empirical Analysis', 18 Strategic Management Journal, 97-109.
- Kan, Steven S. and Hwang, Chun-Sin (1994), *Principles of Economics: Cooperating for Mutual Prosperity and Progress (in Chinese)*, Published by the authors and distributed by Shin Lu Bookstore, Taipei.
- Kan, Steven S. and Hwang, Chun-Sin (1996), 'A Form of Government Organization from the Perspective of Transaction Cost Economics', 7 Constitutional Political Economy, 197-220.
- Kaplan, Steven E. (1989), 'The Effects of Management Buyouts on Operating Performance and Value', 24 Journal of Financial Economics, 217 ff.
- Klausner, Michael and Kahan, Marcel (1995), 'Corporate Contracting: Standardization, Innovation and the Role Contracting Agents', 48 Stanford Law Review.

- Klein (1982), 'The Modern Business Organization: Bargaining Under Constraints', **91** Yale Law Journal, 1521 ff
- Klein, Benjamin (1983), 'Contracting Costs and Residual Claims: The Separation of Ownership and Control', **26** *Journal of Law and Economics*, 367-374.
- Klein, Benjamin (1988), 'Vertical Integration as Organizational Ownership: The Fisher Body-General Motors Relationship Revisited', 4 Journal of Law, Economics, and Organization, 199-213.
- Knight, Frank H. (1921), Risk, Uncertainty, and Profit, New York, Houghton Mifflin Co.
- Knoeber, Charles R. and Flath, David J. (1985), 'On Managerial Shareholding', 34 Journal of Industrial Economics, 93-99.
- Knoeber, Charles R. and Thurman, Walter N. (1994), 'Testing the Theory of Tournaments: An Empirical Analysis of Broiler Production', 12 Journal of Labor Economics, 155-179.
- Kornhauser, Lewis A. and Revesz, Richard L. (1994), 'Multidefendant Settlements under Joint and Several Liability: The Problem of Insolvency', 23 Journal of Legal Studies, 517-542.
- Kraakman, Reinier H. (1984), 'Corporate Liability Strategies and the Costs of Legal Controls', 93 Yale Law Journal, 857-898.
- Kraakman, Reinier H., Park, Hyun and Shavell, Steven (1994), 'When are Shareholder Suits in Shareholders' Interests?', **82** *Georgetown Law Journal*, 1733-1775.
- La Porta, Rafael, Lopez de Silanes, Florencio, Shleifer, Andre and Vishny, Robert W. (1996), *Law and Finance*, National Bureau of Economic Research, No. 5661.
- Lambert, Richard A. and Larcker, D. (1985), 'Golden Parachutes, Executive Decision-making and Shareholder Wealth', 7 Journal of Accounting and Economics, 179-204.
- Lashbrooke, E.C., Jr (1995), 'The Divergence of Corporate Finance and Law in Corporate Governance', 46 South Carolina Law Review, 449-469.
- Lee, D. Scott (1992), 'Management Buyout Proposals and Inside Information', 67 Journal of Finance, 106 ff.
- Lehn, Kenneth (1988), 'Majority-Minority Relationships An Economic View', 13 Canada-United States Law Journal, 135-141.
- Levin, William R. (1985), 'The False Promise of Worker Capitalism: Congress and the Leveraged Employee Stock Ownership Plan', 95 Yale Law Journal, 148-173.
- Lien, Da Hsiang Donald (1990), 'Competition, Regulation and Bribery: A Note', 11 Managerial and Decision Economics, 127-130.
- Lin, Laura (1996), 'The Effectiveness of Outside Directors as a Corporate Governance Mechanism: Theories and Evidence', **90** Northwestern University Law Review.
- Lipton, Martin (1987), 'Corporate Governance in the Age of Finance Corporatism', 136 University of Pennsylvania Law Review, 1-72.
- Lipton, Martin and Rosenblum, Steven A. (1991), 'A New System of Corporate Governance: The Quinquennial Election of Directors', 58 University of Chicago Law Review, 187-253.
- Llebot Majo, Jos Oriol (1996), 'El Sistema de la Responsabilidad de los Administradores, Doctrina y Jurisprudencia (Manager's Liability System, Doctrine and Jurisprudence)', Revista de Derecho de Sociedades.
- Lynk, William J. (1981), 'Regulatory Control of the Membership of Corporate Boards of Directors: The Blue Shield Case', **24** *Journal of Law and Economics*, 159-173.

- Lynk, William J. (1994), 'Property Rights and the Presumptions of Merger Analysis', 39 Antitrust Bulletin, 363 ff.
- Macavoy, Paul W. et al. (1989), Privatization and State-Owned Enterprises: Lesson, Boston, Kluwer Academic Publishers, 360 p.
- Mace, Myles L. (1971), Directors, Myth and Reality, Boston, Harvard Business School.
- Macey, Jonathan R. (1984), 'From Fairness to Contract: The New Direction of the Rules Against Insider Trading', 13 Hofstra Law Review, 9-64.
- Macey, Jonathan R. (1991), 'An Economic Analysis of the Various Rationales for Making shareholders the Exclusive Beneficiaries of Corporate Fiduciary Duties', 21 Stetson Law Review, 23 ff.
- Macey, Jonathan R. and McChesney Fred S. (1985), 'A Theoretical Analysis of Corporate Greenmail', **95** *Yale Law Journal*, 13-61.
- Macey, Jonathan R. and Miller, Geoffrey P. (1987), 'Toward an Interest Group Theory of Delaware Corporate Law', 65 Texas Law Review, 469-523.
- Macey, Jonathan R. and Miller, Geoffrey P. (1991a), 'The Plaintiffs' Attorney's Role in Class Action and Derivative Litigation: Economic Analysis and Recommendations for Reform', 58 University of Chicago Law Review.
- Macey, Jonathan R. and Miller, Geoffrey P. (1991b), 'The Fraud-on-the-Market Theory Revisited', 77 Virginia Law Review, 1001-1016.
- Macey, Jonathan R. and Miller, Geoffrey P. (1995), 'Corporate Governance and Commercial Banking: A Comparative Examination of Germany, Japan and the United States', 48 Stanford Law Review, 73-112.
- Macho-Stadler, I. and Pérez-Castrillo, J. David (1991), 'Double Risque Moral et Delegation (Double Moral Hazzard and Delegation)', **57(3)** Recherches Economiques de Louvain, 277-296.
- Macho-Stadler, I. and Pérez-Castrillo, J. David (1992), 'Le Règle de la Dérogation dans une Rélation d'Agence (Derogation Rule in an Agent-Principal Relationship)', 102-103 Économie et Prévision, 67-77.
- Macho-Stadler, I. and Pérez-Castrillo, J. David (1993), 'Delegar o Centralizar? Qu, dice la Economía de la Información? (What Economics of Information Say about Delegation and Decentralization?)', 52 Cuadernos Económicos de ICE, 25-46.
- Macho-Stadler, I. and Verdier, T. (1989), 'Aspectos Estrategicos de la Separación Accionista-Gestor (Strategic Aspects between Corporate Ownership and Control)', 2(3) De Economía Pública, 25-44.
- Mahoney, Paul G. (1987), 'Mandatory Disclosure as a Solution to Agency Problems', 54 University of Chicago Law Review, 1441-1483.
- Malatesta, Paul H. and Walkling, Ralph A. (1989), 'Poison Pill Securities: Stockholder Wealth, Profitability and Ownership Structure', **25** *Journal of Financial Economics*.
- Manne, Henry G. (1964), 'Some Theoretical Aspects of Share Voting: An Essay in Honor of Merle, Adolf A.', 64 Columbia Law Review, 1427-1444.
- Manne, Henry G. (1965), 'Mergers and the Market for Corporate Control', 73 Journal of Political Economy, 110-120.
- Manne, Henry G. (1966), 'In Defence of Insider Trading', 44 Harvard Business Review, 113-122.
- Mark, Gregory A. (1995), 'Realms of Choice: Finance Capitalism and Corporate Governance. A Review Essay on Mark J. Roe's Strong Managers, Weak Owners', 95 Columbia Law Review, 969-999.

- Marris, R. (1964), The Economic Theory of Managerial Capitalism, New York, Free Press.
- Maug, Ernst (1996), 'Boards of Directors and Capital Structure: Alternative Forms of Corporate Restructuring', 3 Journal of Corporate Finance, April 1997.
- Maug, Ernst (1997), 'Large Shareholders as Monitors: Is There a Trade Off Between Liquidity and Control?', 53 Journal of Finance..
- McChesney Fred S. (1989), 'Economics, Law, and Science in the Corporate Field: A Critique of Eisenberg', 89 Columbia Law Review, 1530-1548.
- Mercuro, Nicholas, Sourbis, Haralambos and Whitney, Gerald (1992), 'Ownership Structure, Value of the Firm and the Bargaining Power of the Manager', 59(2) Southern Economic Journal, 273-283.
- Milgrom, Paul R. (1988), 'Employment Contracts, Influence Activities and Efficient Organization Design', 96 Journal of Political Economy, 517 ff.
- Milgrom, Paul R. and Roberts, John (1990), 'Bargaining Costs, Influence Costs, and the Organization of Economic Activity', in Alt, J. and Shepsle, K. (eds), *Perspectives on Politive Political Economy*, Cambridge, Cambridge University Press.
- Milgrom, Paul R. and Roberts, John (1992), *Economics, Organization and Management*, Englewood Cliffs. NJ. Prentice-Hall.
- Miller, Geoffrey P. and Macey, Jonathan R. (1993), 'Corporate Stakeholders: A Contractual Perspective', 43 University of Toronto Law Journal, 401 ff.
- Miller, Geoffrey P. and Macey, Jonathan R. (1995), 'Corporate Governance and Commercial Banking: A Comparative Examination of Germany, Japan and the United States, 1995, *Stanford Law Review*, 73 ff.
- Mitchell, Lawrence E. (1992), 'A Theoretical and Practical Framework for Enforcing Corporate Constituency Statutes', 70 Texas Law Review, 579-644.
- Mofsky, James and Rubin, Robert D. (1983), 'Introduction: A Symposium on the ALI Corporate Governance Project', 97 University of Miami Law Review, 169-185.
- Moore, Andrew G.T., II (1994), 'Shareholder Rights Still Alive and Well in Delaware: The Derivative Suit: A Death Greatly Exaggerated', **38** Saint Louis University Law Journal, 947-965.
- Murdock, Charles W. (1990), 'The Evolution of Effective Remedies for Minority Shareholders and Its Impact upon Valuation of Minority Shares', 65 Notre Dame Law Review, 425-489.
- Murphy, Kevin J. (1985), 'Corporate Performance and Managerial Renumeration: An Empirical Analysis', 11 Journal of Accounting and Economics, 11-42.
- Newcomb, R. Link (1987), 'The Limitation of Directory Liability: A Proposal for Legislative Reform', 66 Texas Law Review, 411-452.
- O'Connor, Marleen A. (1993), 'How Should we Talk about Fiduciary Duty? Director's Conflict-of-Interest Transactions and the ALI's Principles of Corporate Governance', 61 George Washington Law Review, 954-983.
- Olivella, O. (1990), 'Una Introducción a la Delegación como Mecanismo de Compromiso (An Introduction to Delegation as a Commitment Device)', **7(2)** *Economía Pública*, 37-52.
- Olivella, O. (1995), 'Information Structures and the Delegation of Monitoring', **39** *Annales d'Économie et de Statistique*, 1-32.
- Painter, Richard W. (1994), 'The Moral Interdependence of Corporate Lawyers and Their Clients', 67 Southern California Law Review, 507-584.
- Palmiter, Alan R. (1989), 'Reshaping the Corporate Fiduciary Model: A Director's Duty of

- Independence', 67 Texas Law Review, 1351-1464.
- Portellano, Pedro (1996), Deber de Fidelidad de los Administradores de Sociedades Mercantiles y Oportunidades de Negocio (Fidelity Commitment of Corporation Managers and Bussiness Oportunities), Madrid, Editorial Cívitas.
- Porter, Philip K. and Scully, Gerald W. (1987), 'Economic Efficiency in Cooperatives', 30 Journal of Law and Economics, 489-512.
- Posner, Richard A. (1976), 'The Rights of Creditors of Affiliated Corporations', 43 University of Chicago Law Review, 499-526.
- Pound, John (1989), 'Shareholder Activism and Share Values: The Causes and Consequences of Countersolicitations against Management Antitakeover Proposals', 32 Journal of Law and Economics, 357-379.
- Preite, Disiano (1988), *La Destinazione dei Risultati nei Contratti Associativi* (The Destination of the Results in Partnership Contracts), Milano, Giuffrè, 456 p.
- Radner, Roy (1974), 'A Note of Unanimity of Stockholders' Preferences among Alternative Production Plans a Reformulation of the Ekern-Wilson Model', 5 *Bell Journal of Economics*, 181-184.
- Repetti, James R. (1992), 'Corporate Governance and Stockholder Abdication: Missing Factors in Tax Policy Analysis', 67 Notre Dame Law Review, 971-1035.
- Ribstein, Larry E. (1989), 'Takeover Defenses and the Corporate Contract', **78** Georgetown Law Journal, 17 ff.
- Ribstein, Larry E. (1992a), 'Corporate Political Speech', 49 Washington and Lee Law Review, 109 ff.
- Ribstein, Larry E. (1992b), 'Efficiency, Regulation and Competition: A Comment on Easterbrook and Fischel'. 87 Northwestern University Law Review. 254 ff.
- Ribstein, Larry E. (1993), 'The Mandatory Rules of the ALI Code', **61** George Washington Law Review, 984 ff.
- Roe, Mark J. (1991), 'A Political Theory of American Corporate Finance', 91 Columbia Law Review, 10-67.
- Romano, Roberta (1985), 'Law as a Product: Some Pieces of the Incorporation Puzzle', 1 Journal of Law, Economics, and Organization, 225-283.
- Romano, Roberta (1991), 'The Shareholder Suit: Litigation without Foundation?', 7 Journal of Law, Economics, and Organization, 55-87.
- Romano, Roberta (1992), 'A Guide to Takeovers: Theory, Evidence and Regulation', **9** Yale Journal of Regulation, 119 ff.
- Romano, Roberta (1993), 'Public Pension Fund Activism in Corporate Governance Reconsidered', 93 Columbia Law Review, 795 ff.
- Romano, Roberta (1994), The Genius of American Corporate Law, Washington, DC, The AEI Press.
 Rose-Ackerman, Susan (1991), 'Risk Taking and Ruin: Bankruptcy and Investment Choice', 20
 Journal of Legal Studies, 277-310.
- Ross, Stephen A. (1977), 'The Determination of Financial Structure: the Incentive-Signalling Approach', 8 Bell Journal of Economics, 23-40.
- Ross, Stephen A. et al. (1993), Corporate Finance, Irwin/McGraw-Hill Series in Finance, Insurance, and Real Estate, Homewood/New York, Irwin/McGraw.
- Ruder, David S. (1983), 'Protections for Corporate Shareholders: Are Major Revisions Needed?', 37 University of Miami Law Review, 243 ff.
- Ryan, Patrick J. (1991), 'Strange Bedfellows: Corporate Fiduciaries and the General Law Compliance

- Obligation in Section 2.01(a) of the American Law Institute's Principle of Corporate Governance', **66** Washington Law Review, 413-502.
- Sadat-Keeling, Leila (1984), 'The 1983 Amendments to Shareholder Proposal Rule 14A-8: A Retreat from Corporate Democracy?', 59 Tulane Law Review, 161-198.
- Schleifer, A. and Vishny, Robert W. (1996), A Survey of Corporate Governance, NBER, 5554 Working Paper.
- Schott, Kenneth E. (1983), 'Corporation Law and the American Law Institute Corporate Governance Project', 35 Stanford Law Review, 927-948.
- Schwartz, Alan (1985), 'Products Liability, Corporate Structure, and Bankruptcy: Toxic Substances and the Remote Risk Relationship', 14 Journal of Legal Studies, 689-736.
- Schwartz, Alan (1988), 'The Sole Owner Standard Reviewed', 17 Journal of Legal Studies, 231-235. Schwartz, Donald E. (1984), 'Federalism and Corporate Governance', 45 Ohio State Law Journal, 545-590.
- Seligman, Joel Stock (1986), 'Equal Protection in Shareholder Voting Rights: The One Common Share, One Vote Controversy', **54** *George Washington Law Review*, 687 ff.
- Shavell Steven (1982), 'The Social Versus the Private Incentive to Bring Suit in a Costly Legal System', 11 Journal of Legal Studies, 333 ff.
- Shleifer, Andrei and Vishny, Robert W. (1986), 'Large Shareholders and Corporate Control', 94 Journal of Political Economy, 461-488.
- Shleifer, Andrei and Vishny, Robert W. (1986b), 'Greenmail, White Knights, and Shareholders' Interest', 17 Rand Journal of Economics, 293-309.
- Sidak, J. Gregory and Woodward, Susan (1991), 'Takeover Premiums, Appraisal Rights, and the Price Elasticity of a Firm's Publicly Traded Stock', 25 Georgia Law Review, 783 ff.
- Slagter, Wiek J. (1988), Macht en Onmacht van de Aandeelhouder (Shareholders Lack of Power), Deventer, Kluwer, 47 ff.
- Slagter, Wiek J. (1989), Schaarse Rechten. Afscheidscollege Rotterdam (Scarce Rights), Deventer, Kluwer, 42 ff.
- Smart, Christopher J. (1988), 'Takeover Dangers and Non-Shareholders: Who Should be our Brothers' Keeper?', Columbia Business Law Review, 301-339.
- Smith, Adam (1776), The Wealth of Nations.
- Smith, Gordon G. (1996), 'Corporate Governance and Managerial Incompetence: Lessons From Kmart', **14** North Carolina Law Review, 1037 ff.
- Ståhl, Ingemar (1976), 'Ígande och Makt i Féretagen en Debattinledning, Nationalek Onomiska Féreningens Férhandlingar (Ownership and Power in the Firms An Introduction to Debate, the Economics Association's Negotiations)', 1 Ekonomisk Debatt.
- Stano, Miron (1976), 'Monopoly Power, Ownership Control, and Corporate Performance', 7 Bell Journal of Economics, 672-679.
- Stano, Miron (1978), 'Executive Ownership Interests and Corporate Performance', 42 Southern Economic Journal, 272-278.
- Stiglitz, Joseph E. (1989), 'Principal and Agent', in Eatwell, John, Milgate, Murray and Newman, Peter (eds.) (1989), The New Palgrave: A Dictionary of Economics, London, MacMillan.
- Stout, Lynn A. (1988), 'The Unimportance of Being Efficient: An Economic Analysis of Stock Market Pricing and Securities Regulation', 87 Michigan Law Review, 613-709.
- Summers (1980), 'Worker Participation in the U.S. and West Germany: A Comparative Study from

- an American Perspective', 28 American Journal of Comparative Law, 367 ff.
- Triantis, Alexander J. and George G. (1994), 'Conversion Rights and the Design of Financial Contracts', 72 Washington University Law Ouarterly, 1231-1255.
- Triantis, George G. (1992), 'Secured Debt under Conditions of Imperfect Information', 21 Journal of Legal Studies, 225-258.
- Triantis, George G. (1993), 'A Theory of the Regulation of Debtor-in-Possession Financing', 46 Vanderbilt Law Review, 901-935.
- Triantis, George G. (1994), 'A Free-Cash-Flow Theory of Secured Debt and Creditor Priorities', 80 Virginia Law Review, 2155-2168.
- Triantis, George G. (1995), 'Debt Financing, Corporate Decision Making and Security Design', 26 Canadian Business Journal, 93-105.
- Triantis, George G. (1996), 'The Interplay between Liquidation and Reorganization in Bankruptcy:

 The role of Screens, Gatekeepers and Guillotines', 16 International Review of Law and Economics. 101-119.
- Triantis, George G. and Daniels, Ronald J. (1995), 'The Role of Debt in Interactive Corporate Governance', 83 California Law Review, 1073-1113.
- Turnbull, Shann (1997), Evolution of Business and the Corporate Structure, Corporate Directors' Diploma Course, University of New England, Armidale, Australia, Topic 1.1.
- Utset, Manuel (1995), 'Towards a Bargaining Theory of the Firm', **80** Cornell Law Review, 540 ff. Vanecko, Robert G. (1992), 'Regulations 14A and 13D and the Role of Institutional Investors in Corporate Governance', **87** Northwestern Law Review, 376-421.
- Watts, Ross L. and Zimmerman, Jerold L. (1983), 'Agency Problems, Auditing, and the Theory of the Firm: Some Evidence', **26** *Journal of Law and Economics*, 613-633.
- Weisbach, Michael S. (1988), 'Outside Directors and CEO Turnover', 20 Journal of Financial Economics, 431 ff.
- Weiss, Elliott J. and White, Lawrence J. (1987), 'Of Econometrics and Indeterminacy: A Study of Investors' Reactions to 'Changes' in Corporate Law', 75 California Law Review, 551-607.
- Williamson, Oliver E. (1963), 'Managerial Discretion and Business Behaviour', 53 American Economic Review, 1032-1057.
- Williamson, Oliver E. (1976), 'The Economics of Internal Organization: Exit and Voice in Relation to Markets and Hierarchies', 66 American Economic Review. Papers and Proceedings, 369-377.
- Williamson, Oliver E. (1979), 'Transaction-Cost Economics: The Governance of Contractual Relations', 22 Journal of Law and Economics, 233-261.
- Williamson, Oliver E. (1983), 'Organization Form, Residual Claimants, and Corporate Control', 26 Journal of Law and Economics, 351-366.
- Williamson, Oliver E. (1984), 'Corporate Governance', 93 Yale Law Journal, 1197-1230.
- Winter, Ralph A. (1981), 'Majority Choice and the Objective Function of the Firm under Uncertainty: Note', 12 Bell Journal of Economics, 335-337.
- Winter, Ralph K., Jr (1977), 'State Law, Shareholder Protection, and the Theory of the Corporation', 6 Journal of Legal Studies, 251-292.
- Winter, Ralph K., Jr (1988), 'On Protecting The Ordinary Investor', 63 Washington Law Review, 881-902.
- X. (1996), 'Symposium: The Intersection of Tort and Criminal Law', 76 Boston University Law

Review.

Yablon, Charles M. (1992), 'Book Review of The Corporate Lawyer and Executive Pay: In Search of Excess, by Graef Crystal', 92 Columbia Law Review, 1867-1906.
 Zalecki, Paul H. (1993), 'The Corporate Governance Roles of the Inside and Outside Directors', 24

University of Toledo Law Review.